

# **AGU Advances**

# **RESEARCH ARTICLE**

10.1029/2021AV000515

#### **Key Points:**

- Lake methane emissions increased moving south along a climate and permafrost gradient and were linked with increasing water temperatures
- Lake carbon dioxide exchange decreased moving south and was driven by greater hydrological connectivity and nutrient availability
- Net radiative forcing from the lakes will increase due to increased methane emissions, despite decreasing carbon dioxide emissions

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

M. A. Kuhn, kuhn.mckenzie@gmail.com

#### Citation:

Kuhn, M. A., Thompson, L. M., Winder, J. C., Braga, L. P. P., Tanentzap, A. J., Bastviken, D., & Olefeldt, D. (2021). Opposing effects of climate and permafrost thaw on  $CH_4$  and  $CO_2$  emissions from northern lakes. *AGU Advances*, 2, e2021AV000515. https://doi.org/10.1029/2021AV000515

Received 25 JUN 2021 Accepted 13 OCT 2021

**Peer Review** The peer review history for this article is available as a PDF in the Supporting Information.

#### **Author Contributions:**

**Conceptualization:** Andrew J. Tanentzap, David Bastviken, David Olefeldt **Data curation:** Lauren M. Thompson, Lucas P. P. Braga, Andrew J. Tanentzap, David Olefeldt

Formal analysis: Lauren M. Thompson, Johanna C. Winder, Lucas P. P. Braga, Andrew J. Tanentzap, David Bastviken, David Olefeldt

© 2021. The Authors

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Opposing Effects of Climate and Permafrost Thaw on $\overline{CH}_4$ and $\overline{CO}_2$ Emissions From Northern Lakes

McKenzie A. Kuhn<sup>1,2</sup> <sup>(D)</sup>, Lauren M. Thompson<sup>1</sup> <sup>(D)</sup>, Johanna C. Winder<sup>3</sup> <sup>(D)</sup>, Lucas P. P. Braga<sup>3,4</sup> <sup>(D)</sup>, Andrew J. Tanentzap<sup>3</sup> <sup>(D)</sup>, David Bastviken<sup>5</sup> <sup>(D)</sup>, and David Olefeldt<sup>1</sup> <sup>(D)</sup>

<sup>1</sup>Department of Renewable Resources, University of Alberta, Edmonton, AB, Canada, <sup>2</sup>Department of Earth Sciences and Earth System Research Center, University of New Hampshire, Institute for the Study of Earth, Oceans, and Space, Durham, NH, USA, <sup>3</sup>Ecosystems and Global Change Group, Department of Plant Sciences, University of Cambridge, Cambridge, UK, <sup>4</sup>Department of Biochemistry, Universidade de São Paulo, São Paulo, Brazil, <sup>5</sup>Department of Thematic Studies—Environmental Change, Linköping University, Linköping, Sweden

**Abstract** Small, organic-rich lakes are important sources of methane  $(CH_4)$  and carbon dioxide  $(CO_2)$  to the atmosphere, yet the sensitivity of emissions to climate warming is poorly constrained and potentially influenced by permafrost thaw. Here, we monitored emissions from 20 peatland lakes across a 1,600 km permafrost transect in boreal western Canada. Contrary to expectations, we observed a shift from source to sink of  $CO_2$  for lakes warmer regions, driven by greater primary productivity associated with greater hydrological connectivity to lakes and nutrient availability in the absence of permafrost. Conversely, an 8-fold increase in  $CH_4$  emissions in warmer regions was associated with water temperature and shifts in microbial communities and dominant anaerobic processes. Our results suggest that the net radiative forcing from altered greenhouse gas emissions of northern peatland lakes this century will be dominated by increasing  $CH_4$  emissions and only partially offset by reduced  $CO_2$  emissions.

**Plain Language Summary** Methane  $(CH_4)$  and carbon dioxide  $(CO_2)$  emissions from small lakes are globally significant, yet highly uncertain and our understanding of how  $CH_4$  and  $CO_2$ emissions from small lakes in the north will change with warming and permafrost (i.e., frozen ground) is not well known. Here, we measured emissions from 20 lakes across a 1,600 km climate and permafrost gradient. Our results show that  $CH_4$  and  $CO_2$  emissions followed opposing trends along a north-to-south gradient. We show that increasing  $CH_4$  emissions are strongly associated with warmer temperatures while decreasing  $CO_2$  exchange is linked to shifts in hydrology and within-lake primary productivity brought on by the absence of permafrost. Our results indicate small northern lakes have the potential to be greater sources of  $CH_4$  but also greater  $CO_2$  sinks.

# 1. Introduction

Small lakes emit globally significant amounts of carbon dioxide ( $CO_2$ ) and methane ( $CH_4$ ) to the atmosphere (Hastie et al., 2018; Holgerson & Raymond, 2016; Rosentreter et al., 2021; Wik, Varner, et al., 2016), yet the sensitivity of emissions to climate change is poorly constrained. Fueling the biogeochemistry of small lakes are the relatively warm and organic-rich sediments in shallow waters, and the inflow of nutrients, dissolved gases, terminal electron acceptors, and organic matter from the surrounding landscape (DelSontro et al., 2016; Douglas et al., 2020; Lapierre & Del Giorgio, 2012; Wauthy et al., 2018; Wik, Varner, et al., 2016). Small, shallow lakes with organic-rich sediments are abundant across northern peatlands, which cover ~3% of the global land surface (Hugelius et al., 2020; Messager et al., 2016). Arctic amplification has caused temperatures in northern peatland regions to increase much faster than the global average, and these regions are furthermore experiencing rapid permafrost thaw (Olefeldt et al., 2016; Treat & Jones, 2018). To understand the climate sensitivity of peatland lake  $CO_2$  and  $CH_4$  emissions it is thus necessary to account for both the direct effects of warming and the indirect effects arising from increased hydrological connectivity following permafrost thaw (McKenzie et al., 2021).

Lake  $CH_4$  and  $CO_2$  emissions are both influenced by the direct effects of climate warming on microbial activity. For  $CH_4$ , increasing temperatures influence both  $CH_4$  production (methanogenesis) and oxidation (methanotrophy), but effects on methanogenesis generally dominate and result in high sensitivity of the net



Funding acquisition: Andrew J. Tanentzap, David Olefeldt Investigation: Lauren M. Thompson, Johanna C. Winder, Andrew J. Tanentzap, David Bastviken, David Olefeldt Methodology: Johanna C. Winder, Andrew J. Tanentzap, David Bastviken, David Olefeldt Project Administration: David Olefeldt Resources: Andrew J. Tanentzap, David Olefeldt

Supervision: Andrew J. Tanentzap, David Olefeldt

Visualization: Lauren M. Thompson, Johanna C. Winder, Lucas P. P. Braga, Andrew J. Tanentzap, David Bastviken, David Olefeldt

Writing – original draft: David Olefeldt

Writing – review & editing: Lauren M. Thompson, Johanna C. Winder, Lucas P. P. Braga, Andrew J. Tanentzap, David Bastviken, David Olefeldt  $CH_4$  emissions (Yvon-Durocher et al., 2011, 2014). Temperature sensitivity of net  $CH_4$  emissions is however often assessed within individual ecosystems, for example, seasonally, and it is uncertain whether this sensitivity reflects spatial differences among lakes at broad geographical scales (Morrissey & Livingston, 1992; Rasilo et al., 2015; Sepulveda-Jauregui et al., 2015). Heterotrophic respiration and net emission of  $CO_2$  are also temperature-dependent (Gudasz et al., 2010; Kosten et al., 2010), but relationships between lake net  $CO_2$  emissions and temperatures are often less clear than for  $CH_4$  (Lapierre et al., 2015; Sobek et al., 2005). While most boreal peatland lakes are net sources of  $CO_2$  (Hastie et al., 2018; Lapierre et al., 2015; Weyhenmeyer et al., 2015),  $CO_2$  emissions are influenced not only by respiration but also by  $CO_2$  uptake through primary productivity, which in turn is influenced by nutrient availability and water column light conditions (Lapierre & Del Giorgio, 2012; Pacheco et al., 2014).

Permafrost thaw in peatland regions may have additional direct and indirect effects on lake CO<sub>2</sub> and CH<sub>4</sub> emissions through several processes. Permafrost thaw along lake edges may directly impact GHG emissions through abrupt thaw and the deposition of organic matter into the lakes or through the gentle expansion of saturated lake areas (Turetsky et al., 2020; Walter Anthony et al., 2018). More indirectly, thaw opens both shallow and deep groundwater connections, bringing water to lakes with high concentrations of dissolved organic matter (DOC), nutrients, and weathering products, respectively (Frey & McClelland, 2009; Laudon et al., 2012). Within-lake primary productivity increases with greater nutrient availability, often associated with groundwater inputs but can be inhibited if concentrations of DOM cause substantial light attenuation (Bogard et al., 2020). Greater delivery of CO<sub>2</sub> and CH<sub>4</sub> produced in surrounding organic soils can also cause high lake emissions (Paytan et al., 2015; Weyhenmeyer et al., 2015), while connectivity to deeper groundwater may increase pH and reduce lake CO<sub>2</sub> emissions through pH-driven changes in carbonate equilibrium (Finlay et al., 2015). Higher  $CH_4$  emissions may also be driven by the availability of labile organic matter, supplied from the surrounding landscape, littoral emergent vegetation, or from within-lake primary productivity (Bastviken et al., 2004; DelSontro et al., 2016; Emilson et al., 2018). Conversely, increased delivery of terminal electron acceptors, for example, sulfate and iron, may inhibit methanogenesis (Sivan et al., 2011; Yu et al., 2016). Indirect effects of climate warming through permafrost thaw may thus either enhance or dampen the direct effects of climate warming leading to unclear climate feedbacks from a broader scope than just temperature effects.

We measured CO<sub>2</sub> and CH<sub>4</sub> emissions, alongside water chemistry and microbial community dynamics, during two ice-free seasons from 20 shallow lakes across a 1,600 km permafrost transect within the Interior Plains of western Canada (Figure 1). The Interior Plains of western Canada represents a unique opportunity to use a space-for-time approach to assess climate sensitivity of lake  $CO_2$  and  $CH_4$  emissions, as the region has broadly similar vegetation and a similar glacial history throughout the whole transect. The region is characterized by generally flat topography, sedimentary bedrock overlain by thick, heterogeneous surficial geology (Geological Society of Canada, 2014), and widespread peatlands with abundant lakes (Figure 1 insets). The region has a relatively dry climate throughout, with mean annual precipitation between 300 and 450 mm and mean annual runoff between 50 and 150 mm, but it has a large gradient in mean annual average temperatures (MAAT) between  $-8^{\circ}$ C and  $+2^{\circ}$ C (Table S1 in Supporting Information S1). Peatlands in the Interior Plains started developing shortly after deglaciation ~10,000 years ago, and today have 2-5 m deep peat deposits (Halsey et al., 1998; Hugelius et al., 2020). Permafrost expanded southward after the Holocene thermal maximum and reached its maximum extent following the Little Ice Age, ~1,000 years ago (Pelletier et al., 2017). Permafrost conditions strongly influence peatland hydrology and thus the hydrological connectivity of lakes to their surrounding landscape in this region (Quinton et al., 2019). Widespread permafrost precludes deeper groundwater flow but also blocks shallow flow paths and much peatland runoff enters hydrologically isolated depressions rather than connecting with lakes (Quinton et al., 2019; Vitt et al., 2000). Consequently, with less permafrost in warmer climates, there is increased connectivity of lakes with the surrounding peatlands, while lake position in the landscape determines the potential for regional groundwater recharge or discharge (Hokanson et al., 2019). In this study, we assess the influence of climate and permafrost thaw on CO<sub>2</sub> and CH<sub>4</sub> emissions from peatland lakes throughout boreal western Canada. We estimate the current magnitude of greenhouse gas emissions from lakes in the region and use a spacefor-time approach to evaluate how emissions will change with climate warming.



# **AGU Advances**



**Figure 1.** Locations of study lakes within the Interior Plains of western Canada. The Interior Plains study region is outlined with black lines, and is comprised of the Taiga Plains (north) and Boreal Plains (south, largely permafrost free) ecozones. Permafrost zones, indicated by blue shadings, are from Brown et al. (2002). Five lakes were monitored for seasonal greenhouse gas emission at each study location, indicated by orange circles. The distribution of peatlands within both the study region and the broader circumpolar region is shown in the top right inset, showing that the study region is one of the major northern peatland regions (Olefeldt et al., 2021). Bottom right photos show one lake from each study location, with all lakes surrounded by treed peatlands.

# 2. Methods

### 2.1. Sampling Sites

The studied region includes Taiga Plains and Boreal Plains ecozones of the Interior Plains in boreal western Canada, an area that extends from 70°N to 56°N. This region encompasses the lands of the Metis, Dene Tha', Woodland Cree, Big Stone Cree in Alberta and the lands of the Inuvialuit Settlement Region, Gwich'in Settlement Area, and the Dehcho of the Northwest Territories. The region is characterized by thick peat deposits and generally flat topography (Hugelius et al., 2020; Zoltai, 1993). The region was previously glaciated and is dominated by glacial and glaciolacustrine surficial geologies (Geological Society of Canada, 2014). Across the latitudinal transect, the ecosystems include similar tree species including aspen (*Populus tremuloides*), balsam poplar (*Populus balsamifera*), white spruce (*Picea glauca*), jack pine (*Pinus banksiana*), larch (*Larix laricina*), and black spruce (*Picea mariana*).

We sampled 20 lakes located in four sites within the study region (Figure 1), spanning across different climatic and permafrost zones (Table S1 in Supporting Information S1). The southern maximum extent of permafrost in the region is approximately 100 km south of the sporadic permafrost zone sampling location (Brown et al., 2002). Projected losses of permafrost thaw have the southern maximum extent of permafrost moving north of the sporadic and discontinuous zones by 2100 under RCP 4.5 and complete near-surface permafrost loss is projected for the study region by 2100 under RCP 8.5 (Liu et al., 2021). At each site, we chose five study lakes that had surface areas between 0.3 and 10.1 has (Table S2 in Supporting Information S1), ensuring similar ranges in lake sizes within each location. We only selected lakes surrounded by peatlands and excluded lakes near dirt roads or rivers, which may influence the chemistry of the lakes.



Sampling took place at all sites during the ice-free seasons in 2018 and 2019. We designed the sampling campaigns to capture seasonal patterns at each site from just after ice-out to just before ice-on. Each lake was sampled a minimum of 7–12 times to get a statistically accurate representation of annual diffusive greenhouse gas emissions (Wik, Thornton, et al., 2016). At the northernmost site, each lake was sampled 7–9 times due to logistical limitations of getting to the sites. Exact sampling dates can be found in the available metadata and in Table S3 in Supporting Information S1.

#### 2.2. Water Chemistry

Surface water chemistry was measured from a boat at the location of flux measurements and at the center of the lakes. At each lake, we sampled dissolved  $CH_4$  concentrations using a headspace method (Karlsson et al., 2013) and measured dissolved oxygen, specific conductivity, pH, and temperature in the top 10 cm using a handheld water monitor (Yellow Springs Instrument ProDO). Water samples for DOC and total dissolved nitrogen (TDN) were sampled from the center of the lake following protocols outlined in Burd et al. (2018). Concentrations of DOC andTDN were measured by a TOC-L combustion analyzer with a TNM-L module (Shimadzu, Kyoto, Japan). Specific ultraviolet absorbance (SUVA) at 254 nm was measured from 200 to 700 nm (UV-1280, UV-VIS Spectrophotometer, Shimadzu Corporation, Japan) and was corrected by Milli-Q water blanks. We calculated SUVA by first correcting decadal absorbance at 254 nm (cm<sup>-1</sup>) for interference by Fe as outlined by Weishaar et al. (2003).

Water samples for ion concentrations were filtered and frozen on the day of sampling and kept frozen until analysis. Anions were measured using Ion chromatography (Dionex Ion chromatography DX 600, Thermo Fisher Scientific, US). Cations were measured by ICP-OES (iCAP6300 Duo, Thermo Fisher Scientific, US). Samples for Chlorophyll-*a* analysis were collected using opaque Nalgene bottles. Duplicate samples for each lake were refrigerated and vacuum filtered on the day of collection using 47-mm diameter, ethanol-rinsed GF/F filters (Whatman). The filters were then placed in petri dishes, wrapped in aluminum foil, and were frozen immediately until analyses at the University of Alberta Biogeochemical Analytical Service Laboratory.

#### 2.3. Microbial Sampling

Between July 28 and August 31, 2019, we collected three surface layer ( $\sim$ 15 cm) sediment samples from the edge of each lake where flux measurements were taken. Samples were pooled per lake and DNA was extracted in duplicate from 0.65 g dry sediment per sample using the DNeasy PowerSoil kit (QIAGEN N.V., Hilden, Germany). Sequencing libraries were prepared from 250 ng DNA per sample with the NEBNext Ultra II FS DNA Library Prep kit according to the manufacturer's instructions (New England Biolabs, Ipswich, MA, USA). Samples were sequenced paired end (300 cycles) using a NextSeq 500/550 High Output Kit v2,5 (Illumina, Inc, San Diego, CA, USA). Contamination was negligible, as assessed with both negative (nuclease-free water) and positive (ZymoBIOMICS Microbial Community Standard, Zymo Research Corporation, Irvine, CA, USA) controls. We estimated the abundance of functional marker genes involved in methanogenesis: methyl coenzyme A reductase (mcrA) and methanotrophy: methane monooxygenase subunit A (mmoX) and particulate methane monooxygenase (pmoA), expressed relative to the total number of SILVA 16S hits for each sample with GraftM (Boyd et al., 2018). Whole-metagenome functional profiling relative to the SEED database (used to identify decaheme cytochrome genes and methanogenesis enzyme composition data) was performed with SUPER-FOCUS (Silva et al., 2016), and reads were normalized using the geometric mean of pairwise ratios (Chen et al., 2018). We were unable to extract enough DNA samples from one lake in the discontinuous zone (lake W1) to perform analysis.

### 2.4. Greenhouse Gas Fluxes

Diffusive  $CH_4$  and  $CO_2$  fluxes were measured with small, lightweight chambers (volume = 13.55 L, area = 0.062 m<sup>2</sup>), covered in reflective tape. Diffusive measurements were taken on the edge of the lakes 1–2 m from the shore to avoid littoral vegetation. For lakes that had unstable edges to deploy the chambers from, we placed the chambers on the water surface from an inflatable boat as to not disturb the sediments. In the few cases when sediment disturbance was observed (i.e., change in the linearity of methane flux

and/or higher than normally observed  $CH_4$  concentrations), measurements were stopped and restarted in nearby unaffected locations. We neither avoided nor targeted lake edges with ongoing thermokarst. Most lake edges in the permafrost zones showed evidence of moderate permafrost thaw, but mass wasting of large peat blocks was very uncommon. While there is evidence for both under- and over-estimating lake fluxes based on near shore measurements (Matveev et al., 2016; Schilder et al., 2016), all of the study lakes were surrounded by black spruce trees near the lake edge, providing similarly sheltered wind conditions and comparable environments for comparison across the transect. The chambers were connected to a portable greenhouse gas analyzer (Los Gatos Research, California) in a closed path that recorded the linear change in GHG concentrations within the chamber over time. On each sampling occasion, 3–4 chamber flux measurements of 1–5 min durations (average = 3 min) were taken and then averaged to estimate daytime flux estimate. Typically, each lake was measured on 1–3 days per trip and 7–12 days total. Fluxes were calculated using linear regression (slope) of the change in GHG gas concentration (ppm) inside the chamber over time and the ideal gas law, air temperature, and atmospheric pressure at the time of sampling (Equation 1). Negative slopes for  $CO_2$  measurements were interpreted as  $CO_2$  uptake.  $CH_4$  slopes with  $R^2$  values less than 0.90 were excluded in the daytime average slope calculations.

$$Flux = S * \left(\frac{P * V}{R * T * A}\right) * M * C$$
<sup>(1)</sup>

where *S* is the slope from the linear change in gas concentrations within the chamber over time (atm minute<sup>-1</sup>), *P* is barometric pressure (atm), *R* is the universal gas constant (L atm K<sup>-1</sup> mol<sup>-1</sup>), *V* is the volume of the chamber (L), *T* is the average air temperature inside the chamber (K), *A* is the area of the chamber (m<sup>-2</sup>), *M* is the molar mass of the CH<sub>4</sub> or C-CO<sub>2</sub> (mol mg<sup>-1</sup>) and *C* is a conversion factor to convert from per minute to per day.

We measured ebullitive  $CH_4$  and  $CO_2$  fluxes using a bubble trap (area = 0.073 m<sup>2</sup>) equipped with a syringe and stopcock. Three out of five lakes at each site were equipped with bubble traps, including three traps along the lake edge and three traps in the lake center. Water column depths below the traps ranged from 0.5 m (L5/L1 edges) to 2.5 m (W1/F2 centers; Table S2 in Supporting Information S1). Traps were set at the beginning of each trip and were sampled 2–4 days later. Accumulated gas bubbles were collected manually using syringes. To measure GHG concentrations in the bubbles we reset the traps and disturbed the sediments, immediately collecting 30 ml of sample, stored in 20 ml glass vials sealed with rubber septa (see in Supporting Information S1 for more discussion on ebullition measurements). Samples were analyzed within 2 weeks after collection at the University of Alberta using an SRI-8610C gas chromatograph equipped with an FID detector and methanizer to convert  $CO_2$  into  $CH_4$ . We calculated the ebullitive flux as outlined by Wik et al. (2013).

#### 2.5. Upscaling

We estimated current lake CH<sub>4</sub> and CO<sub>2</sub> emissions from the Taiga Plains by multiplying the area of small and midsized peatland lakes from the Boreal Arctic Wetland and Lake Databset ("BAWLD"; Olefeldt et al., 2021) by the mean CH<sub>4</sub> and CO<sub>2</sub> flux specific to the MAAT reported for each  $0.5 \times 0.5^{\circ}$  grid cell of the database and the number of ice-free days. Daytime diffusive  $CH_4$  emissions were multiplied by a diel correction factor (0.70) to account for diel patterns in  $CH_4$  emissions between day and night (i.e., "24-hr fluxes"; Sieczko et al., 2020). Mean  $CH_4$  and  $CO_2$  emissions and the number of ice-free days were calculated based on linear relationships between mean annual temperature,  $CH_4$  or  $CO_2$  flux, and ice-free days observed across the latitudinal transect (Figure S4 in Supporting Information S1). Total mean  $CH_4$  emission (corrected diffusion + ebullition) was used in upscaling. We used error propagation of the flux uncertainty and ice-cover day uncertainty to calculate the low and high ranges of current and future total annual carbon emissions. Annual carbon emission values do not include lake area uncertainty from BAWLD, which is estimated to be between -21% and +31% for the low-end and high-end confidence bounds (see Section 4 for more discussion on sources of uncertainty). Carbon dioxide emissions were not corrected for nighttime-factor, but evidence suggests diel patterns in CO, respiration are less, bordering negligible for darker, DOC-rich waters (Gómez-Gener et al., 2021), so diel patterns are likely minimal for the lakes that are positive CO<sub>2</sub> emitters. Furthermore, daylight hours across the transect are long, with closer to full days of sunlight throughout



most of the summer period in the northern sites. Therefore, applying the diel factor for daily  $CH_4$  emissions is likely a conservative estimate of daily net  $CH_4$  emissions. Though we do acknowledge that the lakes that experience net uptake of  $CO_2$  during the daytime may be small sinks at nighttime-especially for periods closer to fall and spring. We only scaled across grid cells within the Taiga Plains that were within the temperature MAAT range of the study and not the entire extent of the Interior Plains. See Data Availability for information on the land cover database.

#### 2.6. Radiative Forcing Model

We used a radiative forcing model that applies simple impulse-response functions and assumed the same warming potential timeframes for  $CH_4$  and  $CO_2$  (Joos et al., 2013). We modeled the carbon pools for  $CH_4$ and CO, and mixed atmosphere assumptions following the methods and code by Günther et al. (2020). While CH<sub>4</sub> was modeled as one pool with a simple exponential decay rate, CO<sub>2</sub> was split into five pools with varying decay rates (Frolking et al., 2006) and did not consider the climatic effects of  $CO_2$  from  $CH_4$ oxidation (Myhre et al., 2013). We compared the radiative forcing trajectories under two global temperature change scenarios (RCP 2.6 and RCP 4.5; IPCC's AR5). To account for the effects of Arctic temperature amplification observed across the Northwest Territories, we multiplied the projected changes in average global temperatures for each RCP by three, which is more conservative than the most recent analysis by the Government of the Northwest Territories which suggest that the region could be warming four times faster (2018). We did not include RCP scenario 8.5 because the projected temperature increases are greater than the range of temperatures experienced across the study transect (range =  $9^{\circ}$ C; Table S1 in Supporting Information S1). Prior to 2020, we performed a simplified model spin-up, starting in 1900. In the spin-up, we applied current CO<sub>2</sub> and CH<sub>4</sub> fluxes at a constant rate over a 120-year period (Figure S6 in Supporting Information S1). The purpose of the simplified model spin-up is to bring the model close to an equilibrium point (Turetsky et al., 2020). We then used the year 2020 as a reference point for changes in instantaneous net radiative forcing (Helbig et al., 2017). The model accounts for future changes in  $CO_2$  and  $CH_4$  emissions and the number of ice-free days associated with warming (Figure S5 in Supporting Information S1). The effect of hotspot emissions along thermokarst lake edges was not explicitly accounted for but was implicitly included since our lake edge sampling included moderately affected thermokarst lake edges. Any influence of permafrost thaw on spring ice-out emissions could not be accounted for and is not considered in the model. Uncertainty for the radiative forcing model was calculated based on one standard deviation of the scaling model coefficients, including flux and ice-free day uncertainty.

#### 2.7. Statistical Analysis

All statistical analyses were performed in R statistical software (Version 3.8, R Foundation for Statistical Computing, www.r-project.org). Predictor variables for all models were log-transformed for statistical analyses except water temperature, DOC, and pH. Daytime GHG emission comparisons between zones were performed using mixed-effects models (R Package 3.3.3; Lme4 Package; Bates et al., 2020) with the fixed effect of Permafrost Zone and random effects of the individual lake to account for repeated measures, and season to account for variation in emissions between early, mid, and late-season emissions (Table S4 in Supporting Information S1). Each mixed model for CO<sub>2</sub>, diffusive CH<sub>4</sub>, and ebullitive CH<sub>4</sub> was compared to the respective null model, which only included the random effects, using ANOVA. Individual pairs were then compared using a post-hoc, non-parametric t-test (Bonferroni adjustment; Table S5 in Supporting Information S1). Relationships between environmental variables and GHG emissions were first assessed using a Pearson's correlation matrix ("corrplot" package; Figure S1 in Supporting Information S1). For simple linear regression, structural equation modeling (SEM), and redundancy (RDA) analysis, we used mean values over the entire sampling period and excluded three outlier lakes from analysis in the SEM. The outlier points include lakes L2, L5, and W4 (see Table S2 in Supporting Information S1 for more information on lake IDs). The lakes were designated as outliers due to extreme differences in nutrient concentrations, solute chemistry, and CO<sub>2</sub> emissions, respectively (Figure S7 in Supporting Information S1). We did not include sulfate in the RDA or further analyses based on the correlation plot which indicated that sulfate covaried with the other groundwater proxies, including base cations and conductivity, and did not have a relationship with CH<sub>4</sub> (Figure S1 in Supporting Information S1). The RDA was performed using the vegan





**Figure 2.** Trends in magnitude and variability of lake greenhouse gas emissions across permafrost zones. Emissions of (a) Diffusive  $CO_2$  and (b)  $CH_4$  were measured during the early, mid, and late ice-free season from five lakes each within the continuous (Cont.), discontinuous (Disc.), and sporadic (Spor.) permafrost zones, as well as south of the permafrost region (None). Boxplots indicate the 25th, median, and 75th percentiles of emissions, and the yellow diamonds indicate the average emissions. Different letters above the boxplots for  $CH_4$  indicate statistically significant differences between permafrost zones, accounting for repeated measurements throughout the season (see Section 2 and Table S4 in Supporting Information S1). Diffusive and ebullitive  $CH_4$  flux significant differences are presented separately. Positivie  $CO_2$  fluxes represent a net source to the atmosphere while negative values represent a net sink. Outlier  $CO_2$  fluxes below -800 and above +800 mg C m<sup>-2</sup> d<sup>-1</sup> are not shown. Methane fluxes have not been corrected for diel variation in this figure.

package's rda() function with CO<sub>2</sub> and diffusive CH<sub>4</sub> emission set as the response variables. Methane and CO<sub>2</sub> fluxes were log transformed to normalize the data. Finally, changes to the functional composition of microbial communities were tested using permutational multivariate analysis of variance with vegan's *adonis2* function based on a Euclidean distance matrix.

We used SEMs to quantify the direct and indirect effects of MAAT on environmental variables and average daytime CO<sub>2</sub> and CH<sub>2</sub> emissions (R Package, piecewiseSEM; code adapted from St Pierre et al. (2019). The SEM defines causal links between variables of interest through a system of predefined interconnected linear equations (Shipley, 2009). It then tests the significance of individual paths and the entire model as a whole. The significance of individual paths is determined by a P value < 0.05. However, the model as a whole is defined as a good fit if P > 0.05 from a Fisher's test of directed separation, as this means that the proposed model structure adequately captures the variability in the model variables and the designated paths (Shipley, 2009). Designated model paths must be supported by known links between variables based on prior work. For CO<sub>2</sub> the links and associated references are (a) Chla (as a proxy for productivity) and CO<sub>2</sub> (Engel et al., 2020), (b) phosphate and lake productivity (Carpenter et al., 2001), (c) DOC and productivity (Kissman et al., 2013), and (d) MAAT and phosphate and DOC (Laudon et al., 2012; Roehm et al., 2009; Weyhenmeyer et al., 2015). For CH<sub>4</sub>, the links and prior work included (a) water temperature and CH<sub>4</sub> flux (DelSontro et al., 2016; Yvon-Durocher et al., 2014); (b) water temperature and lake depth (Fang & Stefan, 1999); and (c) MAAT and water temperature (Schneider & Hook, 2010). Iron was not included in the CH<sub>4</sub> SEM because clear links between higher dissolved iron in the continuous permafrost zone could not be established in the prior literature and theories. A summary of the statistical tests used in relation to GHG emissions can be found in Table S6 in Supporting Information S1.

# 3. Results and Discussion

We found that lakes in regions with less permafrost and warmer climates had lower  $CO_2$  emissions, or even  $CO_2$  uptake, and much higher  $CH_4$  emissions, than lakes further north (Figure 2). Lakes in warmer climates also had higher pH, concentrations of base cations, DOC, phosphate (PO<sub>4</sub><sup>3-</sup>), and Chla. Between-lake





**Figure 3.** Influence of lake water chemistry and lake characteristics on greenhouse gas emissions across permafrost zones. (a) Redundancy analysis (RDA) ordination for  $CO_2$  and  $CH_4$  emissions, where explanatory variables include water chemistry variables along with maximum lake depth and mean annual air temperature (MAAT). The circles represent individual lakes, colored by permafrost zonation. Close spatial proximity of the circles to one another indicates that lakes are more similar in their emissions, whereas circles that are far away indicate dissimilarities. The direction of the arrows represents the relationship between the environmental variables and  $CO_2$  and  $CH_4$  flux. Variables with opposite facing arrows are negatively correlated and those with arrows leading in the same direction and orientation are positively correlated. Orthogonally related arrows indicate no relationship. The percent of variability explained by each ordination axis is shown in parentheses. The RDA shows that lake  $CO_2$  and  $CH_4$  fluxes were not correlated, with  $CH_4$  emissions associated with lake depth, water temperature, and MAAT while  $CO_2$  emissions were more closely associated with higher concentrations of nutrients, DOC, and chlorophyll *a*. Base\_Cations represents summed Ca, Mg, and Na concentrations. Potassium (K) is shown separately as it followed a dissimilar trend to the other cations. Structural equation models were constructed individually for (b)  $CO_2$  and (c)  $CH_4$ . Arrows with solid lines and asterisks represent standardized model coefficients, indicating positive and negative relationships. Goodness of fit (Fisher's C values) were 12.90 and 4.85, respectively for the  $CO_2$  and  $CH_4$  models, while P values were 0.40 and 0.30. P values > 0.05 indicate that model structure does not differ from that expected by the data. Outlier lakes were excluded from the RDA and SEM analysis (see Section 2.6).

variability also increased from colder to warmer climates for both  $CO_2$  and  $CH_4$  emissions, and water chemistry (Figure 2). While  $CO_2$  and  $CH_4$  emissions both varied with latitude, they were only weakly related to each other (Figure S2 in Supporting Information S1). Below, we explore the distinct drivers of  $CO_2$  and  $CH_4$ emissions and use the information to project future emissions and the net radiative forcing resulting as feedback to climate warming under different RCP scenarios.

#### 3.1. Connectivity, Primary Productivity, and CO<sub>2</sub> Emissions

Our study shows that climate change is likely to reduce peatland lake  $CO_2$  emissions over the ice-free period, as its positive effects on primary productivity appeared greater than effects on respiration across the transect. We attribute this result primarily to indirect effects of climate through increased hydrological connectivity of lakes to the surrounding landscape. Across the transect, diffusive  $CO_2$  emissions were lower for lakes with higher concentrations of Chl*a*, which also were lakes with higher DOC, light absorbance,  $PO_4^{3-}$ , and pH (Figure 3a). These water chemistry attributes are characteristic of near-surface water in nutrient-rich groundwater-connected peatlands (i.e., fens) that are relatively more widespread in discontinuous and non-permafrost zones (Vitt et al., 2000; Wood et al., 2016; Figure 4). While there was a trend of decreasing  $CO_2$  emissions in warmer climates, the SEM highlighted that this link was mediated through effects on lake water chemistry ( $R^2 = 0.78$ , Fisher's C = 12.9, P = 0.40; P > 0.05 = good fit; Figure 3b). In the SEM, we hypothesized that higher MAAT allows for less permafrost and increased lake connectivity with groundwater-fed peatlands, raising lake DOC, phosphate, and pH, which in turn increases primary productivity, indicated by higher Chl*a* (Engel et al., 2020), and shifts the carbonate equilibrium from  $CO_2$  to bicarbonate, both causing reduced  $CO_2$  emissions. Notably, while we did not design the study to monitor  $CO_2$  emissions from abrupt thaw features, our study does include lake edges impacted by moderate levels





**Figure 4.** Influence of climate and permafrost conditions on greenhouse gas emissions from peatland lakes. Lakes in the continuous permafrost zone were found to be net sources of  $CO_2$  and had low  $CH_4$  emissions, while lakes outside the permafrost region were  $CO_2$  sinks and had high  $CH_4$  emissions. Absence of permafrost allows for increased hydrological connectivity to the surrounding landscape, potentially both deeper flow-paths through mineral soils and near-surface flow-paths through the surrounding peatlands. In this study, we found that lakes outside the permafrost region had generally higher pH and higher concentrations of ions, nutrients, and chlorophyll-*a*-suggesting that elevated primary productivity (PP) outweighs respiration (R) in lakes outside the permafrost region. Lake diffusive  $CH_4$  emissions were primarily linked to lake water temperatures and were 8-times greater south of the permafrost boundary compared to within the continuous permafrost zone.

of thaw. Thus we may have missed small hotspots of higher  $CO_2$  emissions from time-sensitive abrupt thaw locations, however, these features were rare across the transect and we consider our sampling locations to be representative of lake shorelines in the region.  $CO_2$  emission from ebullitive fluxes did not follow a strong north to south trend but emissions were generally higher from the non-permafrost zone than the continuous zone (Figure S3 in Supporting Information S1), suggesting a potential link between temperature and sediment respiration. However, the magnitude of ebullitive  $CO_2$  fluxes was minor compared to diffusive emissions (~7% of the total flux), and did not change the latitudinal trends or net sink/source capacity of individual lakes.

Our findings of decreasing diffusive  $CO_2$  emissions from lakes with higher DOC concentrations contrast most other boreal lake surveys (Lapierre et al., 2015; Weyhenmeyer et al., 2015). Previous studies have shown strong positive relationships between lake DOC concentrations and  $CO_2$  emissions, suggesting that greater connectivity to terrestrial sources of DOC either enhances lake respiration and/or suppresses primary productivity through light attenuation, resulting in higher net  $CO_2$  emissions (Ask et al., 2012; Lapierre et al., 2015). We speculate that the difference between our results and previous studies stems from our inclusion solely of small peatland lakes with relatively high DOC concentrations (range 17–50 mg L<sup>-1</sup>). Other surveys include lakes of various origins and sizes, and thus lakes with much lower DOC concentrations (Ask et al., 2012; Lapierre et al., 2015). For the peatland lakes in this study, it is possible that further elevated DOC concentrations in warmer climates have limited additional influence on respiration, though notably, we did not directly measure respiration or metabolism. However, our results do suggest that higher DOC may promote primary productivity through the delivery of complex nutrients. For example, while the SEM showed a non-significant direct link between DOC and  $CO_2$  flux, there was a significant causal link between DOC and Chla, indicating DOC influences primary productivity and thus indirectly impacts net  $CO_2$  emissions (Figure 3b). Our results suggest that peatland lake  $CO_2$  emissions are likely to respond to climate change in a different way than other lake types with varying DOC and nutrient conditions, such as deeper lakes in regions of Canada and Scandinavia underlain by Precambrian rock, and may also vary regionally among peatland lakes depending on landscape characteristics.

#### 3.2. Temperature Sensitivity, Microbial Dynamics, and CH<sub>4</sub> Emissions

Both ebullitive and diffusive CH<sub>4</sub> emissions were greater in warmer climates, with emissions from lakes in the non-permafrost zone being  $\sim 8$  times greater than emissions from lakes in the continuous permafrost zone throughout the ice-free season. Between-lake variability in  $CH_4$  emissions was also greatest in the sporadic and non-permafrost zones (Figure 2b). Ebullitive emissions represented ~28% of the total C emissions from lakes where it was measured; a fraction that did not vary systematically across the transect. Ebullitive CH<sub>4</sub> emissions were generally highest in spring, while diffusive emissions were highest during summer or fall (Figure S3; Table S5 in Supporting Information S1). Higher MAAT and water temperature were strongly related to higher diffusive CH<sub>4</sub> emissions, while higher base cation concentrations, decreasing lake depth, and lower concentrations of iron were weakly associated with higher emissions (Figure 3a). The lack of a strong relationship between diffusive CH<sub>4</sub> emissions and lake size and depth, as is often found (Holgerson & Raymond, 2016; Wik, Varner, et al., 2016), was not surprising as all studied lakes were relatively small, shallow, and predominately non-stratified (Table S2 in Supporting Information S1). The SEM analysis showed a significant relationship of MAAT with water temperature, which in turn influenced diffusive CH<sub>4</sub> emissions ( $R^2 = 0.67$ , Fischer C = 4.3, P = 0.3; Figure 3c). The apparent temperature sensitivity ( $Q_{10}$ ) of diffusive  $CH_4$  emissions across the transect to changes in water temperature was 16, substantially higher than the  $Q_{10}$  of ~4 that is commonly found for the direct temperature response of  $CH_4$  emissions within individual ecosystems (Yvon-Durocher et al., 2014).

The high apparent temperature sensitivity of  $CH_4$  emissions across the transect suggests the observed direct response to water temperature was amplified by additional factors, possible shifts in microbial communities, redox conditions, or substrate quality (Rasilo et al., 2015). Of these, we found the least support for increasing  $CH_4$  emissions being driven by shifts in substrate quality. Greater lake primary productivity in warmer climates was hypothesized to increase the availability of labile substrate for methanogenesis (Sepulveda-Jauregui et al., 2018), but we found only a weak relationship between  $CH_4$  emissions and Chla (Figure 3a). However, we collected Chla from the lake centers which may not capture primary productivity increase the availability of labile substrates through thermokarst lake expansion. However, visual evidence of ongoing thermokarst lake expansion was found in all permafrost zones, and the highest  $CH_4$  emissions were found in the non-permafrost zone. Lastly, it is possible that shifts in surrounding wetland and littoral vegetation toward denser, taller emergent macrophyte species in warmer climates could provide external sources of labile organic substrates and dissolved  $CH_4$  to the lake (Emilson et al., 2018), although we did not observe any apparent links between shoreline vegetation and  $CH_4$  emissions.

Several lines of evidence instead suggested that shifts in redox conditions and microbial communities contributed to the high climate sensitivity of  $CH_4$  emissions. Our assessment of microbial functional gene abundances showed that warmer climates were associated with a preferential increase of the methanogenesis marker methyl coenzyme A reductase (mcrA) over the methanotrophy markers, methane monooxygenase subunit A (mmoX), and particulate methane monooxygenase (pmoA) (Figure 5). These findings correlate with observations of greater increases in  $CH_4$  production compared to  $CH_4$  consumption in lake sediments with high production rates (D'Ambrosio & Harrison, 2021). While gene abundances do not necessarily correlate with microbial activity (Freitag & Prosser, 2009), this trend was similar to that of observed  $CH_4$  emissions (Figure 2b). A lower abundance of mcrA in the northern lakes may be reflective of a temperature effect, wherein methanogenesis is suppressed under colder conditions. However, we also found evidence suggesting suppression of methanogenesis in the continuous permafrost zone through greater dominance of more thermodynamically favorable iron reduction. Lakes in the continuous permafrost zone trended towards higher concentrations of iron (Fe) in the water (Table S2 in Supporting Information S1), and higher abundances of Fe(III) reduction marker genes (decaheme cytochromes) in sediments (Figure S4 in Supporting Information S1). These conditions could also promote anaerobic  $CH_4$  oxidation (Ettwig





**Figure 5.** Trends in methane-related functional genes among permafrost zones. (a) Relative abundance of the (a) Methanogenesis marker mcrA (methyl coenzyme A reductase) and methanotrophy markers (b) pmoA (particulate methane monooxygenase) increased and (c) mmoX (methane monooxygenase subunit-A) moving south along the transect. Boxplots represent median and quartile ranges for samples taken from each lake once mid-summer 2019 from the continuous (Cont.), discontinuous (Disc.), and sporadic (Spor.) permafrost zones, as well as south of the permafrost region (None). Note the different *y*-axis scales. Differences are not statistically significant, likely due to low sample sizes (n = 5 for every zone except the discontinuous zone where n = 4).

et al., 2016). Higher normalized counts of genes involved in iron acquisition and metabolism in colder climates were furthermore associated with a shift in enzymatic composition from hydrogenotrophic and acetoclastic methanogenesis towards the stoichiometrically less productive methylotrophic form (Rhew et al., 2007; Figure S4 in Supporting Information S1). The mechanisms driving higher dissolved iron and abundances of decaheme cytochromes within the continuous permafrost zone are unclear, but the negative relationship between lake iron, pH, and base cation concentrations suggests that hydrological connectivity to deeper groundwater sources is important (Figure 3a).

Our study suggests that peatland lake  $CH_4$  emissions have a high climate sensitivity, likely driven by increasing  $CH_4$  production in organic-rich sediments. High climate sensitivity of peatland lakes is noteworthy as their current  $CH_4$  emissions are higher than emissions from most other boreal lake types (Kuhn et al., 2021). We note however that the observed  $CH_4$  emissions in this study (average diffusion = 27 mg  $CH_4 m^{-2} d^{-1}$  and ebullition = 11 mg  $CH_4 m^{-2} d^{-1}$ ) were in the low range of what has been reported previously for boreal peatland lakes (diffusive interquartile range  $[IQR] = 9-101 \text{ mg } CH_4 m^{-2} d^{-1}$ , ebullitive IQR = 3-89 mg  $CH_4 m^{-2} d^{-1}$ ; Kuhn et al., 2021). Furthermore, our latitudinal trends in GHG emissions were more pronounced than those found from lakes in other northern regions including other peatland regions (Sepulveda-Jauregui et al., 2015; Serikova et al., 2019). Differences in lake  $CH_4$  emissions between major peatland regions may be due to landscape characteristics and Holocene history which influence water chemistry and organic matter quality, but differences can also be due to climate and measurement methods (including timing, placement, and number of measurements). Our study implements a methodology that constraints ice-free season, diffusive, and ebullitive  $CH_4$  fluxes from a large number of lakes across a permafrost transect–yielding robust findings.

#### 3.3. Future Peatland Lake Emissions and Net Radiative Forcing

We used an atmospheric perturbation radiative forcing model to assess the relative importance of projected shifts in peatland lake  $CO_2$  and  $CH_4$  emissions within the Taiga Plains region until 2100 (Frolking et al., 2006). This model considers the larger radiative forcing efficiency of  $CH_4$  compared to  $CO_2$  (Myhre et al., 2013), and the shorter atmospheric lifetime of  $CH_4$  (Allen et al., 2018). We assumed that emissions and the number of ice-free days will change linearly with increasing temperatures, based on the trends observed across our transect (Figure S5 in Supporting Information S1), and assessed changes in emissions based on the RCP Scenarios 2.6 and 4.5, adjusted for temperature amplification in the Taiga Plains (Figure 6a). This model does not include potential emissions from abrupt permafrost thaw and drastic land slumping, which is projected to have an intense, but short-lived, impact on net GHG emissions from lakes (Walter Anthony





**Figure 6.** Net radiative forcing due to changing  $CO_2$  and  $CH_4$  emissions from Taiga Plains peatland lakes under RCP 2.6 and 4.5 scenarios. (a) The global RCP 2.6 and 4.5 scenarios pathways for the 21st century have been multiplied by a factor of three to account for Arctic amplification in the Taiga Plains, western Canada (Government of the Northwest Territories, 2018). (b) Modeled current and future  $CO_2$  and  $CH_4$  emissions and total emissions ( $CO_2 + CH_4$ ) from peatland lakes on the Taiga Plains. Error bars represent 95% confidence intervals from the model output which includes uncertainty from fluxes and the number of ice-free days. (c) The 21st century radiative forcing resulting from altered lake  $CO_2$  and  $CH_4$  fluxes under the RCP 2.6 and 4.5 scenarios. Radiative forcing is referenced to the year 2020. A scenario of no warming is included, under which the net radiative forcing is only influenced by lake  $CO_2$  emissions while steady  $CH_4$  emissions are in equilibrium with the atmosphere. Radiative forcing due to altered lake  $CO_2$  and  $CH_4$  emissions is expressed as  $10^{-15}$  fW divided by the total lake area of the Taiga Plains. The shaded areas represent the standard deviations from the flux models. Under both the RCP 2.6 and 4.5, we find that the net radiative forcing is dominated by increasing  $CH_4$  emissions and only partly offset by reduced  $CO_2$  emissions.

et al., 2018). Thus, our results may under-estimate net radiative forcing for both  $CH_4$  and  $CO_2$ , however, our study design does take into account emissions from moderately impacted thermokarst edges which were the most common thermokarst feature across the study lakes. Peatland lakes <10 km<sup>2</sup> in size cover ~260,000 km<sup>2</sup> within the Taiga Plains (Olefeldt et al., 2021) and were estimated to emit  $0.20 \pm 0.08$  Tg  $CO_2$ -C  $y^{-1}$  and  $0.048 \pm 0.01$  Tg  $CH_4$ -C  $y^{-1}$  under the current climate. Emissions of  $CO_2$  were estimated to decrease, on average, by 16.5% and 68% under the RCP 2.6 and 4.5 scenarios, respectively (Figure 6b), while  $CH_4$  emissions increased by 31% and 121%, respectively (Figure 6b). Overall, increasing  $CH_4$  emissions dominated the net radiative forcing under both RCP scenarios. Increasing  $CH_4$  emissions added, on average, 5 and 15 fW m<sup>-2</sup> by 2100 under RCP 2.6 and 4.5, respectively, which was only partially offset by reduced  $CO_2$  emissions equivalent to -1 and -2 fW m<sup>-2</sup>, respectively, when compared to the no-warming scenario (Figure 6c).

# 4. Conclusion

Our study shows that climate change is likely to cause reduced  $CO_2$  emissions but greatly increased  $CH_4$  emissions from boreal peatland lakes in western Canada over ice-free periods (Figure 3). Our projections of future peatland lake  $CO_2$  and  $CH_4$  emissions are limited to the ice-free season, and rely on a space-for-time approach with inherent assumptions, but are likely to be robust. Our annual GHG emission estimates contain many uncertainties including measured fluxes and variability in ice-free days across the transect. Further data are required to assess ice-out emissions in the region and associated uncertainties. Several studies have shown that ice-out emissions generally scale with emissions from the ice-free season (Wik et al., 2014), however, ice-out emission studies across latitudinal gradients and within the context of potential changes



in net annual GHG emissions with climate change are still lacking. Additionally, most of the existing iceout emissions are based on comparing water column and CH<sub>4</sub> storage before and after ice-out, and the knowledge about how much of the stored CH<sub>4</sub> is emitted versus oxidized is still limited. Our annual and future GHG estimates do not include uncertainty around the lake area. Additional uncertainty associated with small and midsize peatland lakes in BAWLD is estimated to be between -21% and +31%. We expect the true annual emissions to vary within this range, but it is most likely that small lakes are under-counted (Messager et al., 2016; Olefdt et al., 2021), implying our annual and future estimates are likely conservative. Furthermore, we acknowledge that our measurements represent daytime conditions and that nighttime conditions may lead to higher respiration in some lakes, and may alter the net ice-free estimates of CO<sub>2</sub> exchange. However, our results still highlight key mechanistic differences between CO<sub>2</sub> and CH<sub>4</sub> dynamics in lakes across the study region. Water chemistry and microbial community data allowed for the identification of processes likely responsible for the observed, and opposing CH<sub>4</sub> and CO<sub>2</sub> emissions trends along the transect. We showed that peatland lake  $CO_2$  and  $CH_4$  emissions were driven by different processes and thus not strongly inter-correlated. Both the direct effect of warming and the indirect effect of permafrost thaw was important, where the absence of permafrost in warmer regions allowed for greater variability in lake hydrological connectivity and thus also greater variability in  $CO_2$  and  $CH_4$  emissions. The influence of permafrost thaw on lake productivity and reduced CO<sub>2</sub> emissions, and the extremely high-temperature sensitivity of CH<sub>4</sub> emissions, are likely associated with characteristics of peatland lakes and the hydrology of their surrounding landscape. While additional data from other major peatland regions are needed to support our findings, our study strongly suggests that peatland lakes, which make up  $\sim 18\%$  of all northern lake areas (Olefeldt et al., 2021), need to be considered separately from other lake types when assessing the impact of climate change on future GHG emissions.

# **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

# **Data Availability Statement**

Summary data including coordinates and physical lake characteristics are included in the supplemental (Table S2 in Supporting Information S1). Metadata for water chemistry and greenhouse gas fluxes generated during this study are available at the University of Alberta Libraries' Dataverse network (https://doi. org/10.7939/DVN/LF4WDG). The RCP models for projected temperature change were downloaded from the IAMC AR5 Scenario database (https://secure.iiasa.ac.at/web-apps/ene/AR5DB). Estimates of peatland lake area were extracted from the Boreal Arctic Wetland and Lake Database (BAWLD) and are available at the Arctic Data Center (https://doi.org/10.18739/A2C824F9X). Metagenomic data from this study have been deposited in the European Nucleotide Archive (ENA) at EMBL-EBI under accession number PR-JEB36558 (https://www.ebi.ac.uk/ena/browser/view/PRJEB36558).

#### References

Allen, M. R., Shine, K. P., Fuglestvedt, J. S., Millar, R. J., Cain, M., Frame, D. J., & Macey, A. H. (2018). A solution to the misrepresentations of CO<sub>2</sub>-equivalent emissions of short-lived climate pollutants under ambitious mitigation. *Npj Climate and Atmospheric Science*, 1(1), 1–8. https://doi.org/10.1038/s41612-018-0026-8

Ask, J., Karlsson, J., & Jansson, M. (2012). Net ecosystem production in clear-water and brown-water lakes. *Global Biogeochemical Cycles*, 26(1). https://doi.org/10.1029/2010GB003951

Bastviken, D., Cole, J., Pace, M., & Tranvik, L. (2004). Methane emissions from lakes: Dependence of lake characteristics, two regional assessments, and a global estimate. *Global Biogeochemical Cycles*, 18(4). https://doi.org/10.1029/2004GB002238

Bates, D., Mächler, M., Bolker, B. M., & Walker, S. C. (2020). *lme4: linear mixed-effects models*. Retrieved from https://github.com/lme4/ lme4/

Bogard, M. J., St-Gelais, N. F., Vachon, D., & del Giorgio, P. A. (2020). Patterns of spring/summer open-water metabolism across boreal lakes. *Ecosystems*, 23(8), 1581–1597. https://doi.org/10.1007/s10021-020-00487-7

Boyd, J. A., Woodcroft, B. J., & Tyson, G. W. (2018). GraftM: A tool for scalable, phylogenetically informed classification of genes within metagenomes. Nucleic Acids Research, 46(10), e59–e59. https://doi.org/10.1093/nar/gky174

Brown, J., Ferrians, O., Heginbottom, J., & Melnikov, E. (2002). Circum-Arctic map of permafrost and ground-ice conditions. NSIDC: National Snow and Ice Data Center. https://doi.org/10.7265/skbg-kf16

#### Acknowledgments The authors wish to thank T. Elliot. M.

Schmidt, M. Frederickson, R. Hutchins, and other members of the Catchment and Wetlands Sciences lab group for assistance with field and lab work. The authors thank K. Devito and E. Pugh for providing input on site selection at the Utikima Research Study Area and Carolynn Forsyth for camp facilities at the ArtisInn. Funding for this study was provided by Natural Sciences and Engineering Research Council, Campus Alberta Innovates Program. and the NWT Cumulative Impact Monitoring Program. M. A. Kuhn received support from the Northern Scientific Training Program, University of Alberta and UAlberta North, Vanier Canada Graduate Scholarship, and the W. Garfield Weston Foundation. L. Braga received support from FAPESP 2018/19.247-0 and 2019/24.097-0. J. C. Winder received support from the Bruckmann Fund, Peterhouse College, University of Cambridge, and is grateful to the Molecular Biology Service Unit in the Department of Biological Sciences at the University of Alberta for allowing the use of their lab space. A. J. Tanentzap is funded by H2020 ERC Grant 804673 sEEIngDOM. D. Bastviken was funded by H2020 ERC (Grant 725546, METLAKE), Swedish Research Council VR (Grant 2016-04829), and FORMAS (Grant 2018-01794).

- Burd, K., Tank, S. E., Dion, N., Quinton, W. L., Spence, C., Tanentzap, A. J., & Olefeldt, D. (2018). Seasonal shifts in export of DOC and nutrients from burned and unburned peatland-rich catchments, Northwest Territories, Canada. *Hydrology and Earth System Sciences*, 22(8), 4455–4472. https://doi.org/10.5194/hess-22-4455-2018
- Carpenter, S. R., Cole, J. J., Hodgson, J. R., Kitchell, J. E., Pace, M. L., Bade, D., et al. (2001). Trophic cascades, nutrients, and lake productivity: Whole-lake experiments. *Ecological Monographs*, 71(2), 163–186. https://doi.org/10.1890/0012-9615(2001)071[0163:TCNALP]2.0.CO;2
- Chen, L., Reeve, J., Zhang, L., Huang, S., Wang, X., & Chen, J. (2018). GMPR: A robust normalization method for zero-inflated count data with application to microbiome sequencing data. *PeerJ*, 6(4), 1–20. https://doi.org/10.7717/peerj.4600
- D'Ambrosio, S., & Harrison, J. A. (2021). Methanogenesis exceeds CH<sub>4</sub> consumption in eutrophic lake sediments. *Limnology and Oceanog-raphy Letters*, 6, 173–181. https://doi.org/10.1002/lol2.10192
- DelSontro, T., Boutet, L., St-Pierre, A., del Giorgio, P. A., & Prairie, Y. T. (2016). Methane ebullition and diffusion from northern ponds and lakes regulated by the interaction between temperature and system productivity. *Limnology & Oceanography*, *61*, S62–S77. https://doi.org/10.1002/lno.10335
- Douglas, P. M. J., Gonzalez Moguel, R., Walter Anthony, K. M., Wik, M., Crill, P. M., Dawson, K. S., et al. (2020). Clumped isotopes link older carbon substrates with slower rates of methanogenesis in Northern Lakes. *Geophysical Research Letters*, 47(6), 1–10. https://doi. org/10.1029/2019GL086756
- Emilson, E. J. S., Carson, M. A., Yakimovich, K. M., Osterholz, H., Dittmar, T., Gunn, J. M., et al. (2018). Climate-driven shifts in sediment chemistry enhance methane production in northern lakes. *Nature Communications*, 9(1). https://doi.org/10.1038/s41467-018-04236-2
- Engel, F., Attermeyer, K., & Weyhenmeyer, G. A. (2020). A simplified approach to detect a significant carbon dioxide reduction by phytoplankton in lakes and rivers on a regional and global scale. *Science and Nature*, 107(4), 1–10. https://doi.org/10.1007/s00114-020-01685-y Ettwig, K. F., Zhu, B., Speth, D., Keltjens, J. T., Jetten, M. S. M., & Kartal, B. (2016). Archaea catalyze iron-dependent anaerobic oxida-
- tion of methane. Proceedings of the National Academy of Sciences of the United States of America, 113(45), 12792–12796. https://doi. org/10.1073/pnas.1609534113
- Fang, X., & Stefan, H. G. (1999). Projections of climate change effects on water temperature characteristics of small lakes in the contiguous U.S. Climatic Change, 42(2), 377–412. https://doi.org/10.1023/A:1005431523281
- Finlay, K., Vogt, R. J., Bogard, M. J., Wissel, B., Tutolo, B. M., Simpson, G. L., & Leavitt, P. R. (2015). Decrease in CO<sub>2</sub> efflux from northern hardwater lakes with increasing atmospheric warming. *Nature*, 519(7542), 215–218. https://doi.org/10.1038/nature14172
- Freitag, T. E., & Prosser, J. I. (2009). Correlation of methane production and functional gene transcriptional activity in a peat soil. Applied and Environmental Microbiology, 75(21), 6679–6687. https://doi.org/10.1128/AEM.01021-09
- Frey, K. E., & McClelland, J. W. (2009). Impacts of permafrost degradation on arctic river biogeochemistry. Hydrological Processes, 21, 169–182.
- Frolking, S., Roulet, N., & Fuglestvedt, J. (2006). How northern peatlands influence the Earth's radiative budget: Sustained methane emission versus sustained carbon sequestration. *Journal of Geophysical Research: Biogeosciences*, 111(1), 1–10. https://doi. org/10.1029/2005JG000091
- Geological Society of Canada. (2014). Canadian Geoscience map. https://doi.org/10.4095/295462
- Gómez-Gener, L., Rocher-Ros, G., Battin, T., Cohen, M. J., Dalmagro, H. J., Dinsmore, K. J., et al. (2021). Global carbon dioxide efflux from rivers enhanced by high nocturnal emissions. *Nature Geoscience*, 14(5), 289–294. https://doi.org/10.1038/s41561-021-00722-3
- Government of the Northwest Territories. (2018). 2030 Nwt climate change Strategic Framework. Retrieved from https://www.enr.gov.nt.ca/sites/enr/files/resources/128-climate\_change\_strategic\_framework\_web.pdf
- Gudasz, C., Bastviken, D., Steger, K., Premke, K., Sobek, S., & Tranvik, L. J. (2010). Temperature-controlled organic carbon mineralization in lake sediments. *Nature*, 466(7305), 478–481. https://doi.org/10.1038/nature09186
- Günther, A., Barthelmes, A., Huth, V., Joosten, H., Jurasinski, G., Koebsch, F., & Couwenberg, J. (2020). Prompt rewetting of drained peatlands reduces climate warming despite methane emissions. *Nature Communications*, 11(1), 1–5. https://doi.org/10.1038/ s41467-020-15499-z
- Halsey, L. A., Vitt, D. H., & Bauer, I. (1998). Peatland Initiation during the Holocene in continental western Canada. *Climatic Change*, 40(2), 315–342.
- Hastie, A., Lauerwald, R., Weyhenmeyer, G., Sobek, S., Verpoorter, C., & Regnier, P. (2018). CO<sub>2</sub> evasion from boreal lakes: Revised estimate, drivers of spatial variability, and future projections. *Global Change Biology*, 24(2), 711–728. https://doi.org/10.1023/a:1005425124749
- Helbig, M., Chasmer, L. E., Kljun, N. C., Quinton, W. L., Treat, C. C., & Sonnentag, O. (2017). The positive net radiative greenhouse gas forcing of increasing methane emissions from a thawing boreal forest-wetland landscape. *Global Change Biology*, 23(6), 2413–2427. https://doi.org/10.1111/gcb.13520
- Hokanson, K. J., Mendoza, C. A., & Devito, K. J. (2019). Interactions between regional climate, surficial geology, and topography: Characterizing shallow groundwater systems in subhumid, low-relief landscapes. *Water Resources Research*, 55(1), 284–297. https://doi. org/10.1029/2018WR023934
- Holgerson, M. A., & Raymond, P. A. (2016). Large contribution to inland water CO<sub>2</sub> and CH<sub>4</sub> emissions from very small ponds. Nature Geoscience, 9(3), 222–226. https://doi.org/10.1038/ngeo2654
- Hugelius, G., Loisel, J., Chadburn, S., Jackson, R. B., Jones, M., MacDonald, G., et al. (2020). Large stocks of peatland carbon and nitrogen are vulnerable to permafrost thaw. Proceedings of the National Academy of Sciences of the United States of America, 117(34), 20438– 20446. https://doi.org/10.1073/pnas.1916387117
- Joos, F., Roth, R., Fuglestvedt, J. S., Peters, G. P., Enting, I. G., Von Bloh, W., et al. (2013). Carbon dioxide and climate impulse response functions for the computation of greenhouse gas metrics: A multi-model analysis. *Atmospheric Chemistry and Physics*, 13(5), 2793–2825. https://doi.org/10.5194/acp-13-2793-2013
- Karlsson, J., Giesler, R., Persson, J., & Lundin, E. (2013). High emission of carbon dioxide and methane during ice thaw in high latitude lakes. *Geophysical Research Letters*, 40(6), 1123–1127. https://doi.org/10.1002/grl.50152
- Kissman, C. E. H., Williamson, C. E., Rose, K. C., & Saros, J. E. (2013). Response of phytoplankton in an alpine lake to inputs of dissolved organic matter through nutrient enrichment and trophic forcing. *Limnology & Oceanography*, 58(3), 867–880. https://doi.org/10.4319/ lo.2013.58.3.0867
- Kosten, S., Roland, F., Da Motta Marques, D. M. L., Van Nes, E. H., Mazzeo, N., Sternberg, L. D. S. L., et al. (2010). Climate-dependent CO<sub>2</sub> emissions from lakes. *Global Biogeochemical Cycles*, 24(2). https://doi.org/10.1029/2009GB003618
- Kuhn, M. A., Varner, R. K., Bastviken, D., Crill, P. M., MacIntyre, S., Turetsky, M. R., et al. (2021). BAWLD-CH4: A comprehensive dataset of methane fluxes from boreal and Arctic ecosystems. *Earth Systems Science Data*. https://doi.org/10.5194/essd-2021-141
- Lapierre, J. F., & Del Giorgio, P. A. (2012). Geographical and environmental drivers of regional differences in the lake pCO<sub>2</sub> versus DOC relationship across northern landscapes. *Journal of Geophysical Research: Biogeosciences*, 117(3). https://doi.org/10.1029/2012JG001945

Lapierre, J. F., Seekell, D. A., & del Giorgio, P. A. (2015). Climate and landscape influence on indicators of lake carbon cycling through spatial patterns in dissolved organic carbon. *Global Change Biology*, *21*(12), 4425–4435. https://doi.org/10.1111/gcb.13031

Laudon, H., Buttle, J., Carey, S. K., McDonnell, J., McGuire, K., Seibert, J., et al. (2012). Cross-regional prediction of long-term trajectory of stream water DOC response to climate change. *Geophysical Research Letters*, 39(17), 4–9. https://doi.org/10.1029/2012GL053033

Liu, L., Zhao, D., Wei, J., Zhuang, Q., Gao, X., Zhu, Y., et al. (2021). Permafrost sensitivity to global warming of 1.5°C and 2°C in the Northern Hemisphere. *Environmental Research Letters*, *16*(3), 034038. https://doi.org/10.1088/1748-9326/abd6a8

Matveev, A., Laurion, I., Deshpande, B. N., Bhiry, N., & Vincent, W. F. (2016). High methane emissions from thermokarst lakes in subarctic peatlands. *Limnology & Oceanography*, *61*, S150–S164. https://doi.org/10.1002/lno.10311

McKenzie, J. M., Kurylyk, B. L., Walvoord, M. A., Bense, V. F., Fortier, D., Spence, C., & Grenier, C. (2021). Invited perspective: What lies beneath a changing arctic? *The Cryosphere*, 15(1), 479–484. https://doi.org/10.5194/tc-15-479-2021

Messager, M. L., Lehner, B., Grill, G., Nedeva, I., & Schmitt, O. (2016). Estimating the volume and age of water stored in global lakes using a geo-statistical approach. *Nature Communications*, 7, 1–11. https://doi.org/10.1038/ncomms13603

Morrissey, L. A., & Livingston, G. P. (1992). Methane emissions from Alaska arctic tundra: An assessment of local spatial variability. Journal of Geophysical Research: Atmospheres, 97, 16661–16670. https://doi.org/10.1029/92JD00063

Myhre, G., Shindell, D., Bréon, F.-M., Collins, W., Fuglestvedt, J., Huang, J., et al. (2013). Anthropogenic and Natural radiative forcing: Supplementary material. Climate change 2013: The physical Science Basis. Contribution of working group I to the Fifth assessment Report of the Intergovernmental Panel on climate change: The physical Science Basis. Contribution of working group I to the Fifth assessment Report of, 1–44. Retieved from http://www.climatechange2013.org/images/report/WG1AR5\_Chapter08\_FINAL.pdf%5Cnhttp://www. climatechange2013.org/images/report/WG1AR5\_Ch08SM\_FINAL.pdf

Olefeldt, D., Goswami, S., Grosse, G., Hayes, D., Hugelius, G., Kuhry, P., et al. (2016). Circumpolar distribution and carbon storage of thermokarst landscapes. *Nature Communications*, 7, 1–11. https://doi.org/10.1038/ncomms13043

Olefeldt, D., Hovemyr, M., Kuhn, M. A., Bastviken, D., Bohn, T. J., Connolly, J., et al. (2021). The Boreal-Arctic Wetland and Lake Dataset (BAWLD), *Earth Systems Science Data Discuss*. https://doi.org/10.5194/essd-2021-140

Pacheco, F. S., Roland, F., & Downing, J. A. (2014). Eutrophication reverses whole-lake carbon budgets. *Inland Waters*, 4(1), 41–48. https://doi.org/10.5268/IW-4.1.614

- Paytan, A., Lecher, A. L., Dimova, N., Sparrow, K. J., Garcia-Tigreros Kodovska, F., Murray, J., et al. (2015). Methane transport from the active layer to lakes in the Arctic using Toolik Lake, Alaska, as a case study. Proceedings of the National Academy of Sciences of the United States of America, 112(12), 3636–3640. https://doi.org/10.1073/pnas.1417392112
- Pelletier, N., Talbot, J., Olefeldt, D., Turetsky, M. R., Blodau, C., Sonnentag, O., & Quinton, W. L. (2017). Influence of Holocene Permafrost aggradation and thaw on the paleoecology and carbon storage of a peatland complex in northerwestern Canada. *The Holocene*, 27(9), 1391–1405.

Quinton, W., Berg, A., Braverman, M., Carpino, O., Chasmer, L., Craig, J., et al. (2019). A synthesis of three decades of eco-hydrological research at Scotty Creek, NWT, Canada. *Hydrology and Earth System Sciences*, 23, 2015–2039. https://doi.org/10.1177/0959683617693899

Rasilo, T., Prairie, Y. T., & del Giorgio, P. A. (2015). Large-scale patterns in summer diffusive CH<sub>4</sub> fluxes across boreal lakes, and contribution to diffusive C emissions. *Global Change Biology*, 21(3), 1124–1139. https://doi.org/10.1111/gcb.12741

Rhew, R. C., Teh, Y. A., & Abel, T. (2007). Methyl halide and methane fluxes in the northern Alaskan coastal tundra. Journal of Geophysical Research: Biogeosciences, 112(2), 1–11. https://doi.org/10.1029/2006JG000314

Roehm, C. L., Giesler, R., & Karlsson, J. (2009). Bioavailability of terrestrial organic carbon to lake bacteria: The case of a degrading subarctic permafrost mire complex. *Journal of Geophysical Research*, 114(3), 1–10. https://doi.org/10.1029/2008JG000863

Rosentreter, J. A., Borges, A. V., Deemer, B. R., Holgerson, M. A., Liu, S., Song, C., et al. (2021). Half of global methane emissions come from highly variable aquatic ecosystem sources. *Nature Geoscience*, 14(4), 225–230. https://doi.org/10.1038/s41561-021-00715-2

Schilder, J., Bastviken, D., van Hardenbroek, M., & Heiri, O. (2016). Spatiotemporal patterns in methane flux and gas transfer velocity at low wind speeds: Implications for upscaling studies on small lakes. *Journal of Geophysical Research: Biogeosciences*, 121(6), 1456–1467. https://doi.org/10.1002/2016JG003346

Schneider, P., & Hook, S. J. (2010). Space observations of inland water bodies show rapid surface warming since 1985. *Geophysical Research Letters*, *37*(22). https://doi.org/10.1029/2010GL045059

- Sepulveda-Jauregui, A., Armando, J., Hoyos-Santillan, J., Martinez-Cruz, K., Walter Anthony, K. M., Casper, P., et al. (2018). Eutrophication exacerbates the impact of climate warming on lake methane emission. *The Science of the Total Environment*, 636, 411–419. https:// doi.org/10.1016/j.scitotenv.2018.04.283
- Sepulveda-Jauregui, A., Walter Anthony, K. M., Martinez-Cruz, K., Greene, S., & Thalasso, F. (2015). Methane and carbon dioxide emissions from 40 lakes along a north-south latitudinal transect in Alaska. *Biogeosciences*, 12(11), 3197–3223. https://doi.org/10.5194/ bg-12-3197-2015
- Serikova, S., Pokrovsky, O. S., Laudon, H., Krickov, I. V., Lim, A. G., Manasypov, R. M., & Karlsson, J. (2019). High carbon emissions from thermokarst lakes of Western Siberia. *Nature Communications*, 10(1), 1–7. https://doi.org/10.1038/s41467-019-09592-1

Shipley, B. (2009). Confirmatory path analysis in a generalized multilevel context. *Ecology*, 90(2), 363–368. https://doi.org/10.1890/08-1034.1
Sieczko, A. K., Thanh Duc, N., Schenk, J., Pajala, G., Rudberg, D., Sawakuchi, H. O., & Bastviken, D. (2020). Diel variability of methane emissions from lakes. *Proceedings of the National Academy of Sciences of the United States of America*, 117(35), 21488–21494. https://doi.org/10.1073/pnas.2006024117

Silva, G. G. Z., Green, K. T., Dutilh, B. E., & Edwards, R. A. (2016). SUPER-FOCUS: A tool for agile functional analysis of shotgun metagenomic data. *Bioinformatics*, 32(3), 354–361. https://doi.org/10.1093/bioinformatics/btv584

Sivan, O., Adler, M., Pearson, A., Gelman, F., Bar-Or, I., John, S. G., & Eckert, W. (2011). Geochemical evidence for iron-mediated anaerobic oxidation of methane. *Limnology & Oceanography*, 56(4), 1536–1544. https://doi.org/10.4319/lo.2011.56.4.1536

Sobek, S., Tranvik, L. J., & Cole, J. J. (2005). Temperature independence of carbon dioxide supersaturation in global lakes. Global Biogeochemical Cycles, 19(2). https://doi.org/10.1029/2004GB002264

St Pierre, K. A., Danielsen, B. K., Hermesdorf, L., D'Imperio, L., Iversen, L. L., & Elberling, B. (2019). Drivers of net methane uptake across Greenlandic dry heath tundra landscapes. Soil Biology and Biochemistry, 138(May), 107605. https://doi.org/10.1016/j.soilbio.2019.107605

Treat, C. C., & Jones, M. C. (2018). Near-surface permafrost aggradation in Northern Hemisphere peatlands shows regional and global trends during the past 6000 years. *The Holocene*, *28*(6), 998–1010. https://doi.org/10.1177/0959683617752858

Turetsky, M. R., Abbott, B. W., Jones, M. C., Anthony, K. W., Olefeldt, D., Schuur, E. A. G., et al. (2020). Carbon release through abrupt permafrost thaw. *Nature Geoscience*, 13(2), 138–143. https://doi.org/10.1038/s41561-019-0526-0

Vitt, D. H., Halsey, L. A., & Zoltai, S. C. (2000). The changing landscape of Canada's western boreal forest: The current dynamics of permafrost. Canadian Journal of Forest Research, 30(2), 283–287. https://doi.org/10.1139/cjfr-30-2-283



- Walter Anthony, K. W., von Deimling, T. S., Nitze, I., Frolking, S., Emond, A., Daanen, R., et al. (2018). 21st-century modeled permafrost carbon emissions accelerated by abrupt thaw beneath lakes. *Nature Communications*, 9(1), 1–11. https://doi.org/10.1139/x99-214
- Wauthy, M., Rautio, M., Christoffersen, K. S., Forsström, L., Laurion, I., Mariash, H. L., et al. (2018). Increasing dominance of terrigenous organic matter in circumpolar freshwaters due to permafrost thaw. *Limnology and Oceanography Letters*, *3*(3), 186–198. https://doi. org/10.1002/lol2.10063
- Weishaar, J., Aiken, G., Bergamaschi, B., Fram, M., Fujii, R., & Mopper, K. (2003). Evaluation of specific ultraviolet absorbance as an indicator of the chemical composition and reactivity of dissolved organic carbon. Environmental Science and Technology, 37, 4702–4708.
- Weyhenmeyer, G. A., Kosten, S., Wallin, M. B., Tranvik, L. J., Jeppesen, E., & Roland, F. (2015). Significant fraction of CO<sub>2</sub> emissions from boreal lakes derived from hydrologic inorganic carbon inputs. *Nature Geoscience*, 8(12), 933–936. https://doi.org/10.1021/es030360x
- Wik, M., Crill, P. M., Varner, R. K., & Bastviken, D. (2013). Multiyear measurements of ebullitive methane flux from three subarctic lakes. Journal of Geophysical Research: Biogeosciences, 118(3), 1307–1321. https://doi.org/10.1002/jgrg.20103
- Wik, M., Thornton, B. F., Bastviken, D., Macintyre, S., Varner, R. K., & Crill, P. M. (2014). Energy input is primary controller of methane bubbling in subarctic lakes. *Geophysical Research Letters*, 41, 555–560. https://doi.org/10.1002/2013GL058510
- Wik, M., Thornton, B. F., Bastviken, D., Uhlbäck, J., & Crill, P. M. (2016). Biased sampling of methane release from northern lakes: A problem for extrapolation. *Geophysical Research Letters*, 43(3), 1256–1262. https://doi.org/10.1002/2015GL066501
- Wik, M., Varner, R. K., Anthony, K. W., MacIntyre, S., & Bastviken, D. (2016). Climate-sensitive northern lakes and ponds are critical components of methane release. *Nature Geoscience*, 9(2), 99–105. https://doi.org/10.1038/ngeo2578
- Wood, M. E., Macrae, M. L., Strack, M., Price, J. S., Osko, T. J., & Petrone, R. M. (2016). Spatial variation in nutrient dynamics among five different peatland types in the Alberta oil sands region. *Ecohydrology*, 9(4), 688–699. https://doi.org/10.1002/eco.1667
- Yu, Z. G., Göttlicher, J., Steininger, R., & Knorr, K. H. (2016). Organic sulfur and organic matter redox processes contribute to electron flow in anoxic incubations of peat. *Environmental Chemistry*, 13(5), 816–825. https://doi.org/10.1071/EN15091
- Yvon-Durocher, G., Allen, A. P., Bastviken, D., Conrad, R., Gudasz, C., St-Pierre, A., et al. (2014). Methane fluxes show consistent temperature dependence across microbial to ecosystem scales. *Nature*, 507(7493), 488–491. https://doi.org/10.1038/nature13164
- Yvon-Durocher, G., Montoya, J. M., Woodward, G., Jones, J. I., & Trimmer, M. (2011). Warming increases the proportion of primary production emitted as methane from freshwater mesocosms. *Global Change Biology*, 17(2), 1225–1234. https://doi. org/10.1111/j.1365-2486.2010.02289.x
- Zoltai, S. C. (1993). Cyclic development of permafrost in the peatlands of northwestern Alberta, Canada. Arctic & Alpine Research, 25(3), 240–246. https://doi.org/10.2307/1551820